Author's response to reviews

Title: microRNA-145 promotes differentiation in human urothelial carcinoma through down-regulation of syndecan-1

Authors:

  Tomomi Fujii (fujiit@naramed-u.ac.jp)
  Keiji Shimada (keijish@naramed-u.ac.jp)
  Yoshihiro Tatsumi (tatsumi3@naramed-u.ac.jp)
  Kinta Hatakeyama (kpathol@naramed-u.ac.jp)
  Chiho Obayashi (ohbayashi@naramed-u.ac.jp)
  Kiyohide Fujimoto (kiyokun@naramed-u.ac.jp)
  Noboru Konishi (nkonishi@naramed-u.ac.jp)

Version: 2
Date: 4 June 2015

Author's response to reviews:

June 4, 2015

Dr. Krisha Mae Natan
Journal Editorial Office
BioMed Central

Dear Dr. Natan;

Thank you for suggesting for our submission where we should make changes of our article. We make changes about some parts that are pointed out, and submit the revised manuscript.

As follows,

1.) Line Numbering:

We revise our manuscript to include line and page numbers. We submit our manuscript in Microsoft Word, so we do not insert page breaks in our manuscript.

2.) Title Page:

We include a title page at the front of your manuscript file. It contain authors’ names, institutions, countries and email addresses of all authors, and the full postal address of the submitting author.

Our work will be of special interest to the readers of BMC cancer.

This manuscript has not been published and is not under consideration for publication elsewhere. All authors have contributed to this work. All the authors have read the manuscript and approved this submission. The authors report no conflicts of interest.
Sincerely,

Tomomi Fujii, MD, PhD
Department of Pathology, Nara Medical University School of Medicine
840 Shijo-cho, Kashihara, Nara,
634-8521, Japan
TEL: +81-744-22-3051 ext. 2238
FAX: +81-744-23-5687
E-mail: fujii@naramed-u.ac.jp